Author's response to reviews

Title: Axillary artery to left anterior descending coronary artery bypass with an externally stented graft: A technical report

Authors:

Thanos Athanasiou (tathan5253@aol.com)
Emmanouil I. Kapetanakis (emmanouil.kapetanakis@imperial.ac.uk)
Christopher Rao (christopher.rao@imperial.ac.uk)
Loris Salvador (medloris@yahoo.it)
Ara Darzi (a.darzi@imperial.ac.uk)

Version: 3  Date: 9 February 2008

Author's response to reviews: see over
Dear Mr. Zamvar,

Please accept our response to the two reviews of our above titled manuscript.

First Reviewer: Aris Protopapas

Comments: No Revisions Required

Second Reviewer: George T. Stavridis

Comments: “The relative to the vein graft length of the Dacron tube has not been stated in the text”

Our Response/Revision: We have revised the manuscript accordingly (see highlighted version) adding comments on the relative length of the external conduit to the vein graft in the description of the surgical technique and in the comments sections.

We hope we have adequately addressed the raised issue.

Kind Regards,

Emmanouil Kapetanakis, MD